



# Clinical Commissioning Policy: Dolutegravir for treatment of HIV-1 in adults and adolescents

**Month** 2014

Reference: NHS ENGLAND B06/P/b

## **England NHS England**

### Clinical Commissioning Policy: Dolutegravir for treatment of HIV-1 in adults and adolescents

First published: Subject to Consultation

Prepared by NHS England Clinical Reference Group for HIV

First published Subject to Consultation

Published by NHS England, in electronic format only.

#### NHS England INFORMATION READER BOX

Directorate		
Medical	Operations	Patients and Information
Nursing	Policy	Commissioning Development
Finance	Human Resources	

Publications Gateway F	Reference: 1822	
Document Purpose	Guidance	
Document Name	Clinical Commissioning Policy: Dolutegravir for treatment of HIV- 1 in Adults and Adolescents	
Author	NHS England	
Publication Date	24 June 2014	
Target Audience	Foundation Trust CEs, Medical Directors, Directors of PH, Directors of Nursing, NHS England Regional Directors, NHS England Area Directors, NHS Trust Board Chairs, Directors of Finance, Communications Leads, NHS Trust CEs	
Additional Circulation List	CCG Clinical Leads, CCG Accountable Officers, CSU Managing Directors, Local Authority CEs, GPs	
Description	NHS England will commission Dolutegravir for treatment of HIV-1 in Adults and Adolescents in accordance with the criteria set out in this document. This policy outlines the arrangements for funding this treatment for the population of England	
Cross Reference		
	n/a	
Superseded Docs (if applicable)	n/a	
Action Required		
	Publication on web page for consultation	
Timing / Deadlines (if applicable)		
Contact Details for further information	Specialised Services Team	
	NHS England	
	Floor 5	
	Skipton House	
Document Statu	us	

#### Document Status

This is a controlled document. Whilst this document may be printed, the electronic version posted on the intranet is the controlled copy. Any printed copies of this document are not controlled. As a controlled document, this document should not be saved onto local or network drives but should always be accessed from the intranet

#### **Contents**

Policy Statement	5
Equality Statement	5
Plain Language Summary	5
1. Introduction	6
2. Definitions	6
3. Aim and objectives	
4. Epidemiology and needs assessment	7
5. Evidence base	8
6. Rationale behind the policy statement	
7. Criteria for commissioning	10
8. Patient pathway	11
9. Governance arrangements	11
10. Mechanism for funding	11
11. Audit requirements	11
12. Documents which have informed this policy	12
13. Links to other policies	12
14. Date of review	

#### **Policy Statement**

NHS England will commission dolutegravir for treatment of HIV-1 in adults and adolescents in accordance with the criteria outlined in this document.

In creating this policy NHS England has reviewed this clinical condition and the options for its treatment. It has considered the place of this treatment in current clinical practice, whether scientific research has shown the treatment to be of benefit to patients, (including how any benefit is balanced against possible risks) and whether its use represents the best use of NHS resources.

This policy document outlines the arrangements for funding of this treatment for the population in England.

#### **Equality Statement**

Throughout the production of this document, due regard has been given to eliminate discrimination, harassment and victimisation, to advance equality of opportunity, and to foster good relations between people who share a relevant protected characteristic (as cited in under the Equality Act 2010) and those who do not share it.

#### Plain Language Summary

Dolutegravir is an HIV drug that was approved in Europe in January 2014. This is a relatively new type of HIV drug called an integrase inhibitor (INI). It is the third drug in this class.

HIV treatment usually involves taking three or more drugs in a combination. However, sometimes two, three or four of these drugs are combined in a single pill.

Dolutegravir has the potential to improve care because:

- a) It reduces levels of HIV virus in the body quickly. This is the main aim of HIV treatment.
- b) It causes fewer side effects than other HIV drugs. This includes a much lower risk of common side effects such as mood changes, depression, anxiety, disrupted sleep and suicidal thoughts. Overall, this means the treatment is better tolerated and improves patient safety.

Approximately 10% of HIV positive people cannot use the standard first line treatment (efavirenz) because of these side effects and approximately another 10% change treatment because of them up to two years after starting. These side effects affect quality of life and can reduce adherence, with a risk of drug resistance. A further 10% of patients experience problems with their treatment for whom dolutegravir provides an alternative or even the only option of effective treatment.

The evidence for these benefits came from large randomised studies. Studies looking at switching people on stable therapy are still ongoing.

#### 1. Introduction

HIV treatment has improved greatly over the last two decades and standard of care now involves triple therapy, typically with two nucleos(t)ide reverse transcriptase inhibitors (NRTIs) plus one of the following: a ritonavir-boosted protease inhibitor (PI/r), an NNRTI or an integrase inhibitor (INI) [1].

Effective antiretroviral treatment (ART) requires 95% adherence to drug regimes. Development of new ARV medicines often focuses on improvements in tolerability, reductions in toxicity and drug to drug interactions.

Effectiveness of ART is measured by an undetectable viral load. In 2012, the proportion of patients on treatment with an undetectable viral load was very high: 92% had a viral load of less than 200 copies and 86% had less than 50 copies [2]. Current standard treatment is therefore effective for many people. The availability of generic efavirenz has reduced the cost of standard treatment considerably. New drug treatments need to demonstrate both clinical and cost effectiveness over current standard treatments.

Despite the success of current standard treatment in terms of clinical outcomes, neuropsychiatric side effects have commonly been reported with efavirenz which is currently the most widely prescribed drug. People with a history of psychiatric disorders appear to be at greater risk of serious psychiatric side effects. These may include suicidal ideation and possible increased risk of suicide [3, 4].

Dolutegravir is produced by ViiV Healthcare UK Ltd and is the third HIV integrase strand transferase inhibitor (INI) to be approved. It is the first INI that can be taken as once daily dosing without boosting with a pharmacokinetic enhancer. Dolutegravir received an EU licence for treatment in HIV-1 infected adults and adolescents in January 2014.

HIV drugs are not currently reviewed by NICE to determine their clinical and cost effectiveness.

#### 2. Definitions

The key terms used in this policy and their definitions are:

Antiretroviral therapy (ART): This usually consists of a combination of 3 antiretroviral drugs. A backbone of 2 nucleoside reverse transcriptase inhibitors(NRTI) and a 3rd agent from one of the following classes of drugs: non-nucleoside reverse transcriptase inhibitors (NNRTI), ritonavir boosted protease inhibitors (PI/r) and integrase inhibitors (INI).

**Fixed dose combination** (FDC): Single tablets which combine a number of agents

**First line therapy**: Efavirenz is a recommended first line 3rd agent, given in combination with either tenofovir and emtricitabine or lamivudine and abacavir, and for reasons of clinical effectiveness and cost is the preferred first line option.

NRTI / NNRTI backbone and 3<sup>rd</sup> Agent: These include individual agents often used

in fixed dose combinations including: abacavir and lamivudine; tenoforir and emtricitabine; tenoforir with efavirenz and emtricitabine; tenoforir, rilpivirine and emtricitabine; and tenoforir, elvitegravir, cobicistat and emtricitabine.

**Second line therapy:** The use of alternative 3rd agents where efavirenz cannot be used for reasons of potential or actual intolerance or transmitted HIV drug resistance. Alternative 3rd agents include: the NNRTIs rilpivirine and nevirapine, the INIs raltegravir, elvitegravir/cobicistat and dolutegravir, and the PI/rs darunavir/ritonavir and atazanavir/ritonavir. Drug selection depends on side effects profile, tolerability, resistance profile, drug-drug interactions and cost.

**Viral load:** plasma HIV RNA levels are used to monitor response to antiretroviral therapy. Patients on effective therapy sustain a plasma HIV RNA level of <50 copies/ml (undetectable). Patients who fail to achieve an undetectable viral load or experience a confirmed viral load rebound to above 50 copies/ml are deemed to be experiencing virological failure.

**Intolerance:** patients who are either assessed to be at high risk of adverse effects or experience adverse effects that will or have led to drug discontinuation are deemed to be intolerant.

**Stable patients:** patients who continue to experience an undetectable viral load and who are not experiencing any intolerance to their medication are deemed to be stable.

#### 3. Aim and objectives

This policy aims to identify the evidence and cost implications of routine commissioning of dolutegravir for specific patient groups.

The objectives are to enable access to dolutegravir where its use is supported by clinical evidence and where it is demonstrated to represent good value.

Dolutegravir is price comparative to second line therapies. This policy aims to identify those patients that would benefit from dolutegravir as a second line therapy choice where first line treatment is not clinically indicated, or where patients are experiencing treatment failure or are experiencing intolerance.

#### 4. Epidemiology and needs assessment

The HIV epidemic continues to pose a public health risk in England. By the end of 2012, an estimated 98,400 (CI 93,500-104,300) people were living with HIV in the UK; approximately one in five (21,900, 22% [18%- 27%]) of whom were undiagnosed and unaware of their infection [5]. Whilst HIV-1 remains a lifethreatening disease, effective antiretroviral (ARV) medicines mean that it can be managed as a chronic long term condition. In 2012 there were 71,800 HIV positive

patients in England in HIV services (78,000 in the UK), of whom 61,107 (85.1%) were receiving ART [2]. The annual increase in patients receiving ART between 2011 and 2012 was 4,749 (8.5%). Estimated new patients receiving ART in 2013, 2014 and 2015 are 5,153, 5,591 and 6,166 respectively.

British HIV Association Treatment guidelines currently recommend [1]:

- NRTI backbone: tenofovir and emtricitabine
- Third agent: EITHER atazanavir/ritonavir, OR darunavir /ritonavir, OR efavirenz, OR raltegravir OR elvitegravir/cobicistat

These guidelines remain under review in view of new outcome data, the expiry of patents for standard of care drugs and the availability of new agents.

It is estimated that between 10 - 20 % of patients starting therapy are unable to take efavirenz based regimens and will require initiation on alternative second line therapies in the first 2 years. Up to 10% of patients on treatment also experience treatment failure or resistance requiring use of new or integrase based regimens to achieve viral suppression.

Dolutegravir offers an additional option to all recognised second line therapies.

#### 5. Evidence base

Dolutegravir is the third HIV integrase strand transferase inhibitor (INI) to be made available and has advantages over the other two. In randomised studies it has shown superiority over two other commonly used third agents, efavirenz and boosted darunavir [6, 7]. It has also shown activity against integrase resistant strains of HIV [8]. In summary:

- In antiretroviral naïve patients, dolutegravir has been shown to be non-inferior to raltegravir (Spring 2 study) and superior to tenofovir with efavirenz and emtricitabine, when combined with abacavir and lamivudine (Single study).
- In treatment experienced but integrase naïve patients, dolutegravir has been shown to be superior to raltegravir (Sailing study).
- Dolutegravir has been shown to be superior to boosted darunavir irrespective of the nucleoside backbone that it is combined with (Flamingo study).
- Dolutegravir has been shown to be an effective treatment option in patients with integrase resistant virus (Viking 3 study). The dolutegravir dose in integrase resistance is 50mg twice daily
- In a number of trials, dolutegravir was combined with abacavir/lamivudine including at high viral loads, which was previously a restriction when using abacavir/lamivudine with other drugs.
- Dolutegravir has reduced side effects and improved tolerability compared with current alternatives.

Dolutegravir is a once-daily drug that can be taken with or without food. Treatment adherence is considered to be an important factor in achieving good clinical outcomes and preventing drug-resistance within drug classes. Issues such as tolerability, pill burden, dose frequency, side effects, safety concerns and access to adherence support may impact on a patient's ability to adhere to their treatment regimen.

Several studies have shown higher adherence rates with once daily dosing of ART compared with twice daily [9,10].

Dolutegravir has a good tolerability profile. In phase III studies, approximately 2% of patients stopped the drug due to adverse events compared with 10% taking tenofovir with efavirenz and emtricitabine [6] and 4% taking darunavir/ritonavir [7]. Discontinuation rates were similar compared to raltegravir.

Although the most common first line regimens used in the UK contain efavirenz [9], a proportion of patients are unable to tolerate it due to severe psychiatric side effects that include mood changes, anxiety, depression, sleep disturbance, suicidal ideation and possible increased risk of suicide [3,4].

Dolutegravir has a lower propensity for the development of resistant mutations during treatment compared with raltegravir (11, 12).

Dolutegravir does not require pharmacokinetic boosting which results in complex drug-drug interactions and it does not need to be taken with food. [13].

In the UK, the virological failure rate on current first-line regimens in 2008–2009 was approximately 10% at one year [14]. Around 3% of patients have evidence of triple-class resistance [15]. BHIVA recommend that patients with triple-class resistance switch to a new anti-viral drug regimen containing at least two, and preferably, three fully active agents; an integrase inhibitor is normally required as part of this [1].

Relatively little is known about transmitted integrase resistance as it is not routinely screened for in the treatment naïve population [16], and may only be tested in those failing integrase inhibitor-based therapy.

The proportion of patients who may require a switch from efavirenz ranges from 11% to up to 30% at four years [17,18].

No trial data is currently available on switching to dolutegravir in patients who are currently stable on treatment.

A dolutegravir named-patient programme has been available since 2011 for patients with integrase resistant mutations, and has been accessed by a small number of patients to date.

#### 6. Rationale behind the policy statement

For up to 20% of patients starting their first treatment, it will not be possible to take or continue efavirenz-containing regimens due to resistance, mental health issues, toxicity or intolerance. These patients will require alternative regimens and dolutegravir is at least clinically equivalent to existing alternatives and is broadly of equivalent cost.

Up to 10% of patients on treatment require a new regimen due to treatment failure. Dolutegravir provides another treatment choice and in some cases may be the only option available.

This commissioning policy proposes routine commissioning of dolutegravir for specific patient groups based on evidence that exists to demonstrate superiority and non inferiority compared with some existing therapies and where this would be cost effective to do so.

NHS England has been offered a commercial in confidence discount for dolutegravir. The cost of the drug is comparable to a second line treatment.

#### 7. Criteria for commissioning

Dolutegravir will be routinely commissioned in HIV-1 infected adults and adolescents in the following clinical scenarios:

#### Patients unable to tolerate first line therapy

Patients who are not suitable for or who do not tolerate efavirenz based first line therapy as agreed in the MDT (expected to be no more than 20% in the first two years of treatment). Dolutegravir is a treatment option for this patient group.

Where used, this policy recommends use of dolutegravir with the lowest cost, clinically indicated backbone. This is generally abacavir and lamivudine.

#### Patients failing treatment treatment and those with resistance

Dolutegravir is approved for use in these patients requiring an integrase inhibitor due to recorded treatment failure or resistance:

- In treatment experienced and Integrase inhibitor naïve patients at a dose of 50mg daily.
- In treatment experienced and integrase resistant patients at a dose of 50mg twice daily.

Dolutegravir should be combined with at least two other anti-viral drugs to which the virus is sensitive.

All patients for whom dolutegravir is considered a treatment option for failure and resistance must be considered in an HIV specialist treatment multidisciplinary

(MDT) meeting and the decision of the MDT recorded.

The manufacturer provides dolutegravir to NHS organisations according to the price agreed in confidence with the Department of Health Commercial Medicines Unit and NHS England.

#### **Exclusions**

- Patients switching to dolutegravir who have not been referred to and discussed in the HIV specialist treatment MDT meeting or where the decision about their treatment is not recorded.
- Patients stable on treatment **should not be** switched to dolutegravir. There are no published trial data for switching stable patients on to dolutegravir.
- Use of dolutegravir by providers who are not commissioned by NHS England to provided HIV care and treatment services.
- An increase in the price of dolutegravir would require a review of this policy.

Where clinicians consider switching to dolutegravir for patients not covered in the circumstances above, an Individual Funding Request may be considered where the patient's case in exceptional. The MDT discussion should be included in the IFR.

#### 8. Patient pathway

Commissioned HIV care and treatment providers who meet the service specification initiate and monitor HIV drug treatment. Prescription and monitoring of dolutegravir is in line with the existing patient pathway.

#### 9. Governance arrangements

All patients identified who might benefit from dolutegravir must be referred to and discussed at a specialist HIV MDT and the recommendation recorded. This includes the cohorts identified for routine commissioning as well as any exceptional cases.

#### 10. Mechanism for funding

NHS England is responsible for funding the use of all antiretroviral medicines. Funding for ART is currently on a pass through basis reported to Area Teams. Trusts will be required to separately identify spend on dolutegravir.

#### 11. Audit requirements

All patients considered for treatment with dolutegravir must be referred to and

discussed in, an HIV MDT. Recommendations for treatment must be recorded.

Commissioners will review the audits. This policy will be reviewed by the CRG annually.

#### 12. Documents which have informed this policy

B06/S/a Specialised Human Immunodeficiency Virus (HIV) Services (Adult) – service specification

B06/S/b Specialised Human Immunodeficiency Virus (HIV) Services (Children) – service specification

B06/PS/a Clinical commissioning policy statement: Stribild for the treatment of HIV-1 infection in adults.

#### 13. Links to other policies

This policy follows the principles set out in the ethical framework that govern the commissioning of NHS healthcare and those policies dealing with the approach to experimental treatments and processes for the management of individual funding requests (IFR).

#### 14. Date of review

This policy will be reviewed in April 2016 unless information is received which indicates that the proposed review date should be brought forward or delayed.

#### References:

- 1. Williams I, Churchill D, Anderson J et al. BHIVA guidelines for the treatment of HIV-1-positive adults with antiretroviral therapy 2012 (updated November 2013), HIV Medicine (2014), 15 (Suppl. 1), 1–85
- 2. Public Health England Personal Communication (SOPHID 2012)
- 3. Mollan K et al. Hazard of suicidality in patients randomly assigned to efavirenz for initial treatment of HIV-1: a cross-study analysis conducted by the AIDS Clinical Trials Group (ACTG). abstract 670, ID Week, San Francisco, California October 2-6, 2013
- 4. Summary of product characteristics Efavirenz (25 June 2013). Date accessed 28/1/14 via www.medicines.org.uk
- 5. Aghaizu A, Brown AE, Nardone A, Gill ON, Delpech VC & contributors. HIV in the United Kingdom 2013 Report: data to end 2012. November 2013. Public Health England, London accessed Dec 13).
- Walmsley S, Antela A, Clumeck N et al. Dolutegravir plus abacavirlamivudine for the treatment of HIV-1 infection. N Engl J Med. 2013 Nov 7;369(19):1807-18

- 7. Feinberg J, Bonaventura C, Khuong M-A et al. Once-daily dolutegravir (DTG) is superior to darunavir/ritonavir (DRV/r) in antiretroviral-naive adults: 48 week results from FLAMINGO (ING114915). Abstract h-1464a. 53rd Interscience Conference on Antimicrobial Agents and Chemotherapy. Denver, Colorado. September 10-13, 2013
- 8. Nichols G, Mills A, Grossberg R et al. Antiviral activity of dolutegravir in subjects with failure on an integrase inhibitor-based regimen: week 24 phase 3 results from VIKING-3. J Int AIDS Soc 2012; 15(Suppl 4). http://dx.doi.org.10.7448/IAS15.6.18112
- 9. Buscher A, Hartman C, Kallen MA et al.Impact of antiretroviral dosing frequency and pill burden on adherence among newly diagnosed, HAART naive HIV patients (Int J STD AIDS 2012; 23(5):351-355
- 10. Nachega JB, Parienti JJ et al. Effect of once daily dosing and lower pill burden antiretroviral regimens for HIV infection: A Meta-analysis of Randomised Controlled Trials. Abstract PS4/5 14th European AIDS conference October 16th-19th Brussels, Belgium.
- 11. Raffi F, Rachlis A, Shellbrink H-J et al. Once daily Dolutegravir verses Raltegravir in antiretroviral naïve adults with HIV-1 infection Lancet 2013; 381 735-43
- 12. Cahn P, Pozniak A, Mingrone H et al. Dolutegravir versus raltegravir in antiretroviral-experienced integrase-inhibitor-naive adults with HIV: week 48 results from the randomised, double-blind, non-inferiority SAILING study. Lancet 2013; 382:700-708
- 13. Summary of product characteristics, Stribild 150 mg/150 mg/200 mg/245 mg film-coated tablets. Gilead Sciences Ltd. Date accessed 7/1/14 via www.medicines.org.uk
- 14. Pillay D, Dunn D. UK HIV Drug resistance database. Annual Report 2008/09. Available at http://www.ctu.mrc.ac.uk/pdf/HIV Res Report09.pdf).
- 15. UK HIV Drug Resistance Database. Prevalence of HIV drug resistance in ART-naïve patients by calendar year. 2007. Available at www.hivrdb.org.uk
- 16. Asboe D et al, British HIV Association guidelines for the routine investigation and monitoring of adult HIV-1-infected individuals 2011. HIV Medicine (2012), 13, 1–44),
- 17. Daar ES et al. Atazanavir Plus Ritonavir or Efavirenz as Part of a 3-Drug Regimen for Initial Treatment of HIV-1 A Randomized Trial. Ann Intern Med. 2011;154:445-456.
- 18. Rockstroh JK, DeJesus E, Lennox JL et al. Durable Efficacy and Safety of Raltegravir Versus Efavirenz When Combined With Tenofovir/Emtricitabine inTreatment-Naive HIV-1-Infected Patients: Final 5-YearResults From STARTMRK. Acquir Immune Defic Syndr 2013; 63: 77-85